Author's response to reviews

Title: REFLEX: a social-cognitive group treatment to improve insight in schizophrenia study protocol of a multi-center RCT

Authors:

Marieke GHM Pijnenborg (g.h.m.pijnenborg@rug.nl)
Mark M Van der Gaag (m.van.der.gaag@psy.vu.nl)
Claudi CLH Bockting (c.l.h.bockting@rug.nl)
Lisette EM Van der Meer (l.van.der.meer@med.umcg.nl)
Andre A Aleman (a.aleman@med.umcg.nl)

Version: 2 Date: 12 July 2011

Author's response to reviews:

Dear Editor,

Herewith I am resubmitting manuscript 1051229581562381, entitled REFLEX: a social cognitive treatment to improve insight in schizophrenia: study protocol of a multi-center RCT. I have added the requested trial registration details. The documents with regard to medical ethical approval were emailed to the editorial office.

The study was funded by a Euryi grant of the European Science Foundation (NWO044035001), which is not a commercial organisation. Thus, there is no conflict of interests to report.

Currently, there are no other manuscript based on this study protocol under submission with other journals.

The area of interest is 'Psychiatry' I tried to specify this under the heading 'manuscript details', but this did not work.

Hopefully, I have provided all requested details.

Yours Sincerly, marieke pijnenborg